\documentclass[12pt]{article}
\usepackage{graphics}\usepackage{amsmath}
\newcommand{\course}{Math 222A}
\newcommand{\quarter}{W03}
\newcommand{\handout}{C}
\newcommand{\hno}{C}
\input{/h1/fa/baker/include/preamble}

\input{/h1/fa/baker/include/kcommands.tex}
\begin{document}


\chapter{Homework}

\noindent Days of the course will be described as  1-M, 1-W, 1-F, 2-M, etc.

\vspace{12pt}

\noindent Due 1-F:  Problems B-1, B-2, B-5, B-7.

\noindent (This means problems from the last section of Handout B.  Future
assignments will have six to eight problems.  Problems will
generally be a mixture of easier ones to build intuition and
harder ones to develop theory.)
\end{document}
